Author’s response to reviews

Title: Effective Feedback to Improve Primary Care Prescribing Safety (EFIPPS) a pragmatic three-arm-cluster randomised trial: designing the intervention (clinicaltrials.gov registration NCT01602705)

Authors:

Karen N Barnett (karen.barnett@ed.ac.uk)
Marion Bennie (marion.bennie@strath.ac.uk)
Shaun Treweek (streweek@mac.com)
Chris Robertson (chris.robertson@strath.ac.uk)
Dennis J Petrie (dennis.petrie@unimelb.edu.au)
Lewis D Ritchie (l.d.ritchie@abdn.ac.uk)
Bruce Guthrie (B.Guthrie@dundee.ac.uk)

Version: 2 Date: 25 March 2014

Author’s response to reviews:

Dear Editor

Many thanks for the opportunity to resubmit our manuscript. This paper describes the development of an intervention for a clinical trial and therefore we found it more difficult to fit neatly into the standard format. I have amended the extensive method section that described, in detail, each of the intervention components. This section has now been split into Methods and Results to reflect the development of the components (methods) and the final output of each component (results). I hope that you find these changes satisfactory. Thank you again for considering our manuscript for publication in Implementation Science.

Yours sincerely

Dr Karen Barnett